Accelerator Status Report

Run Schedule: Jan. 3, 2011 through May 13, 2011
March 9, 2011
Injector    Energy 61.76 MeV
· Since Hall C has been down and Hall A only took low current beam. The injector needs fine tuning for high current before Hall C is back online. Reza is scheduled to do it sometime today. The 0L07 spectrometer is on again.
· The injector was good last week delivering beam to Hall A.

· The injector has been alright. Sometimes beam orbits do not reproduce after the gun HV off. The injector steering script should be used to restore the orbits.

· (2/25) The laser spot was moved to boost the QE. 

Accelerator   Linac Energy 549.00 MeV
· Separators for pass 1 and pass 4 are on and beam was steered to the dumplettes of Hall C and Hall A. A is pass 4 and C pass 1. 2T line aperture is tight horizontally and we are going to address it today.
· (3/8) Maintenance day. 0L07 spectrometer on; 6E02 viewer replaced; Arc9A earth leakage sensor replaced; PSS maintenance and many other things (48 items were scheduled. See entry # 1573258 for details.).
· Last week some multiharp data were collected to improve QWeak beam quality.

· The optics was stable but beam availability is about 50% due to the power interruptions and a CHL trip.

· (3/1) Did some beam studies trying to improve beam quality for Hall C. 

· (3/1) ESR tripped during a cable checkup (1.85 hrs.). 

· (2/26) The two way 1-pass split was tried but not successful. Box supplies do not regulate well at low current.

· (2/25) Around 0420 there was power interruption caused by Virginia Power maintenance work outside the Lab. After most systems were recovered VBV2L22B would not open. The valve would not open because the power cord was left disconnected during the vacuum work in the region. The valve was jumpered open and the jumper needs to be removed since the power cable was connected. We lost more than 21 hours. 
Hall A---Pass 4, E08 - 010
· (3/9) 4-pass separation was done during owl shift. The hall will not be ready for swing shift. The hall line has not been setup yet.
· (3/8) The finished the current run. Switching to experiment E07-006. The vacuum in the Compton region has been degrading. The hall does not need the Compton.
· Hall A beam is fine.

· (2/26) Hall A lost its target chamber vacuum during movement of the BigBite spectrometer. The RadCon rope used to define a rad area was attached to the spectrometer and the vacuum line. During the movement the vacuum line was damaged. Remember when   
· (2/25) Friday we magnetically separated 1-pass beam into Hall A since both B and C were down. The hall has been taking low current (<50uA) and the hall has been happy with beam. Hall A beam size is kind of large (more than 300x300 micrometers) but for what they are doing the size is an issue. 
Hall B ---Pass 5, TPE
· The hall has been down since Friday preparing a new experiment (HDice).
Hall C--- Pass 1, QWeak 
· The hall is still working on its vacuum and will not be ready for beam until sometime tomorrow.  
· Hall C has been down since Friday due to the Qtor power supply. The power supply was damaged when power was interrupted Friday. 

